Author's response to reviews

Title: Postgraduate education for Chinese medicine practitioners: a Hong Kong perspective

Authors:

Vincent CH Chung (vchung@cuhk.edu.hk)
Michelle PM Law (pmlaw@cuhk.edu.hk)
Samuel YS Wong (yeungshanwong@cuhk.edu.hk)
Stewart W Mercer (sm83z@clinmed.gla.ac.uk)
Sian M Griffiths (siangriffiths@cuhk.edu.hk)

Version: 4 Date: 9 February 2009

Author's response to reviews:

Title - Please use sentence case in the title, both in the manuscript file and the submission system i.e. remove all unnecessary capitals (capitalise only the first word, and proper nouns, and no full stops/periods). For example you would change "The Title of My Prestigious Publication: A Scientific Study" to "The title of my prestigious publication: a scientific study"

1. Changed in both resubmission and in the submission system.

Please remove the numbers proceeding the authors' names

2. Numbers removed.

Qualifications & Job titles - Please remove the authors' qualifications (e.g. PhD / MD) and job titles from the manuscript and submissions system

3. Information deleted

Please remove the horizontal lines above and below the author list.

4. Lines Removed

'Submitted as Original Research
BMC Medical Education, Jan 2009 '

Please remove this text from the title page.

5. Text removed

CHINA -> China

6. Changed

Word-counts, Table counts, and Running title Ä¢â€œ Please remove these.
7. Information deleted
Highlighting/tracking- Please remove all highlighting and tracking from the manuscript.

8. Highlighting / tracking removed
Please remove the manuscript title from the head of the Abstract

9. Title removed
Abstract - Abstracts must be structured into Background, Methods, Results, Conclusions. Please remember to also update the Abstract details on the submission page.

10. Changes made in both paper and submission system

Manuscript sections - Manuscript sections should include (in the following order): Abstract; Background; Methods; Results; Discussion; Conclusions; Abbreviations (if any); Competing interests; Authors' contributions; Acknowledgements; References; Figure legends (if any); Tables (if any); Description of Additional files (if any).

Introduction -> Background

11. Changes made
Please move the tables from the main body of the manuscript text and locate them after the reference list.

12. Changes made
References - The reference list should contain all authors' names, if there are 30 or below. The term "et al." should not be used

13. Amendments Made
References - References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example "text [1]."

14. Amendments Made
Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance.

15. Reference 35 has been changed.